
Daily Status Mon 2012-Oct-23
Weather:
Nice crisp and cool morning.  Temps hovering around 8C with humidities in the upper 60%.  

Status:
All stations are communicating via bluetooth and wifi.

Station 6 down on arrival by Cockpit.  Had trouble logging into the dsm but it finally logged me in.  I did a 'adn' then 'aup' to see if I could get the system 
talking again. Extremely slloooooowwwwww communications with Ah6.  The 'aup' did not work.  I do see sensors alive in Minicom though.  Giving Ah6 a 
reboot to see if things begin to work.

I did see if the head of the M21, 10m sonic was loose but it was not.  

Gave a tour to EOL Admins today.  

I do want to add that the base trailer has been neglected this project and I am very dissapointed that it got to this condition.  We all need to do a better job 
in getting organized and vacuuming the carpet.  This will make the trailer last a lot longer and make attitudes a lot better.

Sensors and Issues:
All sensors look good but Ah6 due to no data coming down the pipeline.

To Dos:
Ah6 needs help.  Fixed trh's in base.  Try to figure out what probes that are found laying around that are good.  Organize base for show-and-tell today and 
tomorrow.
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